Title: Influence of antioxidant (L-ascorbic acid) on tolbutamide induced hypoglycaemia/antihyperglycaemia in normal and diabetic rats

Authors:
Satyanarayana Sreemantula (nandinisai@hotmail.com)
Eswar K Kilari (ekilari@rediffmail.com)
VishnuVardhan A (vishani_pharma@rediffmail.com)
Rajasekhar Jaladi (rajasekhar_27@hotmail.com)

Version: 6 Date: 1 March 2005

Author's response to reviews: see over
From:
Dr. Satyanarayana Sreemantula
Professor
University College of Pharmaceutical Sciences
Andhra University, Visakhapatnam.

To,

Alison Z Shaw, Ph.D.
Assistant Editor
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
LONDON W1T 4LB
UK

Dear Dr Shaw,

I am herewith submitting our revised manuscript (MS 1365507333385237). The following are the answers to the questions raised by you

Q1. Affiliation - Please include the country in the affiliation.

The country name have been included in the affiliation

Q2. Corresponding author - Please state that the asterisk indicates the corresponding author.

Yes, the asterisk indicates the corresponding author.

Q3. Competing interests - Please include a 'Competing interests' section below the Conclusions (see below for details). If there are none to declare, please write 'The authors declare that they have no competing interests'.

'The authors declare that they have no competing interests'.

Your early response for publication in BMC Endocrine Disorders is highly appreciated.

Thanking you

Yours sincerely,

Satyanarayana Sreemantula